Author’s response to reviews

Title: Arterial dP/dtmax accurately reflects left ventricular contractility during shock when adequate vascular filling is achieved.

Authors:

Philippe Morimont (ph.morimont@chu.ulg.ac.be)
Bernard Lambermont (b.lambermont@chu.ulg.ac.be)
Thomas Desaive (tdesaive@ulg.ac.be)
Nathalie Janssen (njanssen@chu.ulg.ac.be)
Geoffrey Chase (geoff.chase@canterbury.ac.nz)
Vincent D'Orio (V.DOrio@ulg.ac.be)

Version: 3 Date: 17 February 2012

Author’s response to reviews:

Authors’ contributions (added)
PM conceived the study, participated in the experiments, analyzed the data and drafted the manuscript. BL and TD participated in the experiments, analyzed the data and helped to draft the manuscript. NJ participated in the experiments. GC participated in the design of the study and helped to draft the manuscript. VD participated in the design and coordination of the study and helped to draft the manuscript. All authors read and approved the final version of the manuscript.